Author’s response to reviews

Title: The Research and Evaluation of Antipsychotic Treatment in Community Behavioral Health Organizations, Outcomes (REACH-OUT) Study: Real-World Clinical Practice in Schizophrenia

Authors:

Kruti Joshi (kjoshi33@its.jnj.com)
Lian Mao (LMao@its.jnj.com)
David Biondi (DBiondi@its.jnj.com)
Robert Millet (rmillet@carolinabehavioralcare.com)

Version: 1 Date: 27 Oct 2016

Author’s response to reviews:

Per Anna Clark's decision letter we have made the following revisions, which were necessary prior to the manuscript being sent to peer reviewers:

1. We have uploaded a STROBE checklist.

2. We have added the trial registration date to the end of the abstract.

3. We have updated the IRB/ethics information.

Kind regards,

Kruti Joshi